Author's response to reviews

Title: Academic Leagues: a Brazilian way to teach about cancer in medical universities.

Authors:

Diogo AV Ferreira (dii.valente@gmail.com)
Renata N Aranha (renataaranhauerj@yahoo.com.br)
Maria HF Ornellas (mariahelenaornellas@gmail.com)

Version: 4 Date: 11 October 2015

Author's response to reviews:

dear editors

We present a new version of the manuscript considering the requested modifications.

1. The name of the Ethics Committee was incorporated into the text.

2. Interviews were not used in this study. Data were obtained through gray literature review

3. Legends and Table 1 were included at the end of the references.

4. We confirm that all authors leveram and approved the final manuscript (information included in the Authors contributions).

5. English is written under review by Edanz Group Ltd